Author’s response to reviews

Title: Postoperative remote lung injury and its impact on surgical outcome

Authors:
Lin Chen (linchen17@hust.edu.cn)
Hailin Zhao (hailin.zhao06@imperial.ac.uk)
Azeem Alam (azeemalam1993@yahoo.com)
Emma Mi (ezm20@cam.ac.uk)
Shiori Eguchi (110102ms@gmail.com)
Shanglong Yao (ysltian@163.com)
Daqing Ma (d.ma@imperial.ac.uk)

Version: 2 Date: 06 Feb 2019

Author’s response to reviews:

1. Manuscript body formatting.

- Rename "Introduction" into Background.

Response: We have renamed Introduction into Background.

2. Funding.

Please state clearly the role the funder(s) had in your study in the "funding" section of the declarations

Response: The funders have no role in regarding study design, data collection, analysis and manuscript preparation. We have made this point clear in the Funding section.
3. Declarations

Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section and Abbreviations, and be before the References

Ethics approval and consent to participate
Consent for publication
Availability of data and material
Competing interests

Funding

Authors' contributions

Acknowledgements

If the information required is already provided in the main manuscript, please also copy the relevant statements to the Declarations.

If any of the sections are not relevant to your manuscript, please include the heading and write 'Not applicable' for that section.

Response: All sections listed above were included in the section of Declarations.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: The uploaded manuscript is clean and free of comments and tracked changes.